Author's response to reviews

Title: Dissemination of Strongyloides stercoralis in a patient with systemic lupus erythematosus after initiation of Albendazole: a case report

Authors:

Catherine J Hunter (cathie.hunter@excite.com)
Mikael Petrosyan (mpetrosyan@chla.usc.edu)
Morris Asch (chunter@chla.usc.edu)

Version: 2 Date: 17 March 2008

Dear Editor(s):

I am hereby submitting the revised copy of “Dissemination of Strongyloides stercoralis in a patient with systemic lupus erythematosus after initiation of Albendazole: a case report” for your consideration. We appreciate the reviewer’s comments and have included changes within the manuscript to reflect their recommendations. A response to reviewer’s comments is included below. Should you have any questions please do not hesitate to contact me.

Yours truly,

Catherine Hunter, M.D.

Response to reviewers

Response to Mark Pietroni:

Thank you for your comments.
Comment 1: Modification of title and abstract to include the diagnosis of SLE:
Response: This change has been implemented.

Comment 2: Inclusion of drug doses in the abstract.
Response: These modifications have been made in accordance with the recommendations.

Comment 3: Confusion regarding timing of albendazole therapy.
Response: This has been modified to accurately reflect the timing of events.

Response to Ning Tang

Comment: “Valuable case report, but need to lay stress on completion and accuracy of laboratorial and clinical data collection for strengthening validity, credibility and reliability of case report(s).”
Response: Thank you for your comments. We assure the reviewer and editor that all data is factual and correct. We are unsure of what particular modifications are specifically required/ suggested by Dr. Tang. We hope that the modifications that we have made are sufficient to meet the standards and requirements for publication. Should any further additions be required, please do not hesitate to contact me.